Author's response to reviews

Title: White cell count and platelet count associate with histological alcoholic hepatitis in jaundiced harmful drinkers

Authors:

Timothy Hardy (timothy.hardy@ncl.ac.uk)
Christopher Wells (chris.wells@nth.nhs.uk)
Stuart Kendrick (stuart.kendrick@ncl.ac.uk)
Mark Hudson (Mark.Hudson@nuth.nhs.uk)
Christopher P Day (Chris.day@ncl.ac.uk)
Alistair Burt (alistair.burt@ncl.ac.uk)
Steven Masson (steven.masson@nuth.nhs.uk)
Stephen F Stewart (sstewart@mater.ie)

Version: 2 Date: 20 August 2012

Author's response to reviews:

- Ethics statement: Please document within the methods section of your manuscript the specific name of the organization that granted ethical approval to your study going ahead.

Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

- Consent Statement

As you may be aware, it is a requirement to have written informed consent from all patients for publication and we will not proceed with your manuscript until you have confirmed this. The manuscript should therefore include a statement to this effect in the Acknowledgements section, as follows: "Written consent for publication was obtained from the patient or their relative."

An appropriate statement has been made at end of the methods section:
Ethics for the original antioxidant trial was granted by Newcastle and North Tyneside Local Ethics Committee [12]; research was conducted in compliance with the Helsinki Declaration.